Author’s response to reviews

Title: Trust and Respect in the Patient-Clinician Relationship Preliminary Development of a New Scale

Authors:

Paul Crits-Christoph (crits@pennmedicine.upenn.edu)
Agnes Rieger (arieger@wellesley.edu)
Averi Gaines (againes2013@gmail.com)
Mary Connolly Gibbons (gibbonsm@pennmedicine.upenn.edu)

Version: 2 Date: 09 Oct 2019

Author’s response to reviews:

1. Acknowledgements

Acknowledgements has been added.

2. Form of consent to participate

Written consent obtained. Now noted.

3. Role of funding body

Now specified.

4. All authors read and approved

Statement included.

5. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.